Author’s response to reviews

Title: Infarcted Warthin tumor with mucoepidermoid carcinoma-like metaplasia: a case report and literature review

Authors:

Kenji Yorita (kenjiyorita@gmail.com)
Hideyuki Nakagawa (hktokunaka1119@yahoo.co.jp)
Katsushi Miyazaki (katsushi@bronze.ocn.ne.jp)
Junya Fukuda (fukudaj5@gmail.com)
Satoshi Ito (s-itou@kfy.biglobe.ne.jp)
Makoto Kosai (mkosai@tenor.ocn.ne.jp)

Version: 2 Date: 30 Nov 2018

Author’s response to reviews:

The case report should include past medical, social, environmental, family and employment history.

Our response: Thank you for your suggestions. We added his past medical, social, environmental, and family history in the revised manuscript. However, his employment history could not be available.

Give detailed physical and neurological examination on admission. What was the pulse, blood pressure and temperature, on admission?

Our response: Thank you for your suggestions. We added his vital signs and physical and neurological examination on admission. He had a medical history of hypertension, and high blood pressure was obtained on admission. Tenderness in the area of the right parotid gland was the only abnormal finding of physical and neurological examination. Although we previously wrote that he had no history of smoking in the first manuscript, he had a smoking history. His smoking history is corrected in the re-revised manuscript.
Give all results of laboratory findings (i.e. CBC, liver and renal functions), urinalysis, serology, microbiology etc

Our response: Thank you for your suggestions. We added the data of laboratory test including CBC, serological test, and urinary test.

Give information about follow-up for at least 6 months.

Our response: We could extend follow-up period to eight months because we medically checked him on November 8, 2018.